

Import competition from China and unemployment. An analysis using Spanish workers'micro-data.

Vicente Donoso Víctor Martín Asier Minondo **WP 05/14**

ICEI Working**papers**

Abstract

In the period 1997-2011, import competition from China multiplied by five in the Spanish manufacturing sector. In this paper we analyze whether this severe increase in import competition from China is associated with a higher probability of becoming unemployed in the Spanish manufacturing sector. Linking industry-level data on imports with the working histories of 141,000 manufacturing workers, we show that import competition from China is positively associated with the probability of becoming unemployed. In particular, a standard deviation increase in import competition from China raises the probability of becoming unemployed between 0.8 and 3.5 percentage points, which represents between a 9% and a 44% increase relative to the unconditional probability of becoming unemployed. In contrast, we do not find any effect of import competition from China on manufacturing wages. Also, our estimations show that there is weak evidence of a positive association between a higher import competition from China and the probability of switching to an employment outside the manufacturing sector.

Keywords: imports, China, Spain, unemployment, skills, manufacturing

JEL Classification: F16, J23

© Vicente Donoso, Victor Martín y Asier Minondo. Vicente Donoso, Universidad Complutense de Madrid e ICEI Victor Martín, Universidad Rey Juan Carlos e ICEI Asier Minondo, Universidad de Deusto e ICEI.

Acknowledgements: We would like to thank Francisco Requena, Paulo Guimarães and participants in the Research Seminar at Universidad Complutense de Madrid and at XVI Encuentro de Economía Aplicada in Granada for very valuable comments and suggestions. The authors acknowledge financial support from the Complutense Institute for International Studies (ICEI) of the Complutense University of Madrid, the Spanish Ministry of Economy and Competitiveness (MINECO ECO2010-21643 and ECO2011-27619, co-financed with FEDER, and MINECO HAR2010-18544), and from the Basque Government Department of Education, Language policy and Culture.



Índex

1. introduction	7
2. Database and descriptive analysis	9
3. Ecoometric analysis	13
3.1 Imports from China and the probability of becoming unemployed	13
3.2 Technological progress, the financial crisis and alternative measures of trade exposure.	17
4. The impact of import competition from China on wages and sectoral reallocation	18
5. Conclusions and policy implications	22
6. References	24

1. Introduction

One of the salient features of the current globalization process is the emergence of China as a major exporter. During the period 1997-2011 the share of Chinese exports in total world merchandise exports multiplied by more than three (from 3.3% to 11.1%). The increase in Chinese exports was particularly intense in manufacturing, where the share rose from 3.7% in 1997 to 14.4% in 2011.¹

This large increase in import competition from China has aroused concerns regarding its impact on the demand for labor in highwage countries' manufacturing industries. These concerns are not new. During the 1990s, different studies started to analyze the impact of low-wage countries' exports on high-wage countries' labor markets (Krugman and Lawrence, 1994; Wood, 1994). They used aggregated data at the industry level and differentiated between broad labor categories, such as skilled and unskilled workers. These studies, with few exceptions, concluded that import competition from low-wage countries had a tiny contribution to the decline in the share of unskilled workers in manufacturing; in contrast, technological change, biased against unskilled workers, seemed to play a much larger role in this process (Berman et al. 1998).

However, the surge of Chinese exports during the 2000s motivated a series of analyses aiming to assess the robustness of the conclusions reached by the first generation of studies (Krugman, 2008). This reassessment coincided with the access to datasets that provide rich micro-level data on firms and workers. Studies using these new datasets show that import competition from China had a sizable negative effect on high-wage countries' manufacturing firm survival and on worker labor market outcomes. However, they also conclude that there is heterogeneity in how firms and workers respond to this competition. Using US firm-level data, Bernard et al. (2006) show that plant survival and growth are negatively associated with industry exposure to low-wage country exports. Nevertheless, these authors also show that within an industry the impact of competition from low-wage countries has a lower impact on capital-intensive and skill-intensive plants. Regarding heterogeneity across workers, Ebenstein et al. (2012a) show that occupation characteristics, such as routineness, play a decisive role in determining the impact of Chinese competition on individual wages, even for workers with the same educational level.

The contribution of this paper is to provide new evidence on the still limited literature that analyzes the impact of Chinese import competition on high-wage countries' labor market outcomes, controlling for heterogeneity both at the firm and the individual worker level. Taking Spain as a case study, this paper provides evidence on the effect of Chinese import competition on a European country characterized by a rigid labor market, where demand shocks are mostly absorbed by changes in the quantity of workers, rather than wages. Moreover, the wage level in Spain is lower than in countries such as the US, leading to a productive specialization which is more intensive in unskilled labor than in the US, and hence, at a higher risk of being negatively influenced by a surge of imports from China.

To perform this study, we link industrylevel data on imports with a dataset that comprises the working histories of around 141,000 Spanish manufacturing workers. The database identifies the worker and the firm in which the worker is employed, which allows to control for observed, and nonobserved, heterogeneity both at the firm and the individual worker level. We show that during the period 1997-2011, a larger import competition from China is positively associated with a higher probability of becoming unemployed. In 5particular, a standard deviation increase in import competition from China raises the probability of becoming unemployed between 0.8 and 3.5 percentage points, which represents between a 9% and 44% increase relative to the unconditional probability of becoming unemployed. In contrast, we do not find any impact of import competition from China on manufacturing wages, while the evidence of a positive association between a higher import competition from China and the probability of switching to an employment outside the manufacturing sector is weak.

¹ The figures were calculated from the World Trade Organization database, available from www.wto.org.

This paper is related to a large literature that has analyzed the impact of low-wage countries' import competition on high-wage countries' labor markets. As mentioned above, during the 1990s, different studies used aggregated industry-level data to assess the impact of low-wage countries' imports on the demand for labor in high-wage countries (Krugman and Lawrence, 1994; Wood, 1995; Leamer, 1998). During this decade we also find the first studies that analyzed the impact of offshoring to low-wage countries on high-wage countries' labor-market outcomes (Feenstra and Hanson, 1996 and 1999).

In the next decade, the access to detailed firm-level and worker-level data has allowed a finer analysis on the impact of lowwage countries' import competition on highwage countries' labor market outcomes. As mentioned before, Bernard et al. (2006) find that US manufacturing plant survival and growth are negatively associated with exposure to low-wage countries' imports, although the impact differed across firms. Mion and Zhu (2013) show that import competition from China had a large negative impact on the employment of Belgium low-tech firms. They also find that competition from China induces a skill-upgrading among Belgian firms. Bloom et al. (2012) find that import competition from China induces technical change in high-wage countries.

Harrison and McMillan (2011) find that, in general, offshoring to low-wage countries substitute for domestic employment in US manufacturing firms. Using data on Danish firms and workers, Hummels et al. (2011) find that offshoring tends to increase the highskilled wage and decrease the low-skilled wage. They also find that low-skill workers suffer more from the displacement effects of offshoring. Using a similar dataset as the one used in this paper, Ebenstein et al. (2012a) find that import competition and offshoring to China is associated with wage declines, especially for workers performing routine tasks in the US. Iacovone et al. (2013) analyze the impact of import competition from China on a middleincome country labor market. Using detailed plant-level data from Mexican manufacturing firms, they conclude that import competition

from China raises the probability to exit the market for small firms but has no significant effect on large firms.²

Rather than using firm-level or individual worker data, Autor et al. (2013) exploit the differences in manufacturing specialization across US local labor markets to analyze the impact of import competition from China. They find that local labor markets specialized in industries where the increase in Chinese imports was large experienced a stronger decline in labor force participation, higher unemployment levels, and reduced wages. Using a similar methodology, Donoso et al. (2014) show that Spanish provinces specialized in industries with higher exposure to Chinese imports also experienced large declines in manufacturing employment.

This paper is also related to the literature that uses longitudinal data on workers' labor-history to analyze the factors that determine the transition from employment to unemployment in Spain. Previous studies have shown that having a short-term contract, being a young foreign fe-male worker and commanding a low education level increase the probability of transitioning from employment to unemployment (Medina et al., 2010; Montero and Regil, 2011; Díez-Catalán and Villanueva, 2012).

The rest of the paper is organized as follows. Section 2 presents the database and some descriptive statistics. Section 3 presents the results of the econometric estimations on the relationship between import competition from China and the probability of becoming unemployed. Section 4 analyzes the effect of Chinese competition on other labor market variables such as wages and sectoral reallocation. Section 5 concludes and provides some 2 Some studies using workers flows data analyze the impact of total imports, rather than imports from China, on employment and wages. Using German workers flow data, Görg and Görlich (2012) do not find any significan effect of imports on manufacturing employment and wages; however, Lurweg and Udhe (2010) show that trade has a negative, although small, effect on employment and wages in trade-sensitive manufacturing industries. Geishecker and Görg (2008) find that outsourcing reduces the wage for low-skill workers but increases the wage of high-skill workers; Bachmann and Braun (2011) find that the hazard of transitioning to unemployment rises with international outsourcing for medium-skilled and older workers. Using Austrian workers' flows data, Egger et al. (2007) find that an increase in outsourcing reduces the probability of staying in, or changing into, the manufacturing sector.

policy implications.

2. Database and descriptive analyses

The main dataset used for our empirical analysis comes from the Continuous Sample of Working Lives (Muestra Continua de Vidas Laborales, hereinafter MCVL), provided by the Spanish Ministry of Labor and Social Security. MCVL is a micro-level dataset built upon Spanish administrative records. By means of a simple random sampling system, it consists of a representative sample (4% - 1.2 million individuals) of the population registered with the Social Security administration over the sampling year. The sample includes workers, but also pension earners and recipients of unemployment benefits. The first MCVL wave was carried out in 2004. Since this year the MVCL has a proper longitudinal design. Once a worker is randomly selected by a MVCL wave, she will be included in subsequent waves as long as she remains registered in the Social Security. To ensure representativeness, the individuals that cease their relationship with the Social Security are substituted by new members. The MVCL contains the work histories of all individuals in the sample back to 1967.

The MCVL provides several pieces of information suited for our research. Some information is related to individual characteristics, such as age, gender, and nationality; other information is related to the job-contract and the nature of the job, such as occupation, experience in the firm, whether the worker has a full-time or a part-time job, and whether the contract has discounts regarding social security costs; contract information also documents the cause, if any, for the termination of the contract. Finally, the dataset also provides information of employers' characteristics, such as economic activity, firm size and location.

We pool data from the MVCL 2004, 2005, 2006, 2007, 2008, 2009, 2010 and 2011 waves. We select individuals, aged between 16-65 years, with at least one record as employees with a permanent, or an open-ended, contract along the period 1997-2011 in the manufacturing sector. We exclude self-employed workers and salaried workers with a temporary contract due to the difficulty in de-

termining the real cause of the termination of their job records. In any case, employees with permanent contracts represent almost 70% of total employment in the manufacturing sector. Workers should be enrolled in the general regime of the social security administration to be included in the sample.³

The selection of the period 1997-2011 is determined by two reasons. First, trade and production data to measure the competitive pressure from Chinese imports is only available from the mid-1990s onwards. Second, as mentioned before, the MVCL is only representative of the population registered with the Social Security administration for the sampling year. Although the MVCL has retrospective information, the representativeness of this information declines as we go back in time. Bearing this limitation in mind, we only step back until 1997. From 2008 onwards, the Spanish economy suffered a strong negative shock due to the financial crisis and the burst of the housing bubble. We include the first four years of the great recession period in our analysis (2008-2011). However, in the sensitivity section, we show that our results are not driven by the effects of the Great Recession on the Spanish labor market.

For each year, we classify work spells into two distinct categories: work spells that finish and the worker becomes unemployed; and work spells that remain alive. The first category excludes situations in which the contract ends due to the choice of the worker, retirement, sick leave or maternity leave. We do not consider that a contract ceases if the worker becomes unemployed, but she is employed by the same firm in a period no longer than 15 days.⁴

After applying all the above mentioned filters, and deleting records with missing data, our sample consists of a total of 1,031,275 observations from 140,647 individuals. The number of observations by year and the distribution of individuals by personal characteristics

³ More than 80% of workers in Spain are enrolled in the general regime. Some workers, due to their occupation, are enrolled in different regimes, such as agricultural workers, fishermen, miners, or self-employed workers.

⁴ In our sample, there are workers that have more than one labor contract alive. They represent 2.2% of the individuals included in the sample.

are presented in Table 1. It shows that most individuals in the sample are males (76%), have an age between 30 and 44 years (45%), and are occupied in activities that demand few skills (67%).

Table 2 shows the unconditional probabilities of becoming unemployed. As average, the probability of becoming unemployed for a salaried worker with a permanent contract in the manufacturing sector during the period Previous studies have analyzed the probability of transitioning from employment to unemployment in Spain using longitudinal data (Medina et al., 2010; Montero and Regil, 2012). These studies also find that males have a lower probability of becoming unemployed than women. However, contrary to our results, they find that young people have a higher probability of becoming unemployed than older people. This difference is explained because these studies include workers with temporary

Table 1. Number of observations and distribution by personal and occupational characteristics.

	Number of	Mala		Age (%)		Occ	up Skill	(%)
Year	observations	(%)	16-29	30-44	45-65	High	Medium	Low
1997	51,203	80.3	19.9	43.8	36.5	12.9	20.0	67.1
1998	56,904	79.8	20.9	42.7	36.6	12.9	19.7	67.4
1999	64,709	78.6	22.8	42.0	35.3	12.7	19.7	67.6
2000	69,786	78.0	24.1	41.3	34.6	12.5	19.5	68.0
2001	73,340	77.2	24.1	41.7	34.2	12.5	19.6	67.9
2002	74,566	76.5	24.2	42.7	33.1	12.8	20.1	67.1
2003	75,121	76.2	23.4	43.3	33.3	12.8	20.2	67.0
2004	75,677	75.7	22.2	44.3	33.5	13.0	20.2	66.8
2005	76,004	75.4	21.2	45.3	33.5	12.8	20.2	67.0
2006	75,591	74.9	20.1	46.4	33.5	12.6	19.8	67.6
2007	75,747	74.4	18.8	47.4	33.8	12.9	20.0	67.1
2008	73,282	74.2	17.1	48.2	34.7	13.2	20.2	66.6
2009	68,391	74.3	14.8	49.1	36.1	13.6	20.5	65.9
2010	62,639	74.4	12.8	49.9	37.3	14.1	20.8	65.1
2011	58,148	74.4	10.7	50.1	39.2	14.4	20.9	64.7
Total	1,031,275	76.2	20.0	45.2	34.8	13.0	20.1	66.9

Source: authors' calculations based on MCVL.

1997-2011 is 8.1 percent. Since the start of the Great Recession there is an increase in the unconditional probability of becoming unemployed. In particular, in 2008 the probability rises to 10.1%, in 2009 to 12.6%, and then declines to 9.5% in 2010 and 9.1% in 2011. Table 2 also shows that during the period of analysis the probability of becoming unemployed is larger for females, older workers and workers performing tasks that require low or medium skills.

contracts, which are widely used when hiring young people. As the cost of firing is much low-er in temporary than open-ended contracts, the probability of becoming unemployed is much larger for younger workers. As in our study, Díez-Catalán and Villanueva (2012) also find that workers performing high-skill tasks have a lower probability of becoming unem-ployed.

The competitive pressure of Chinese import is proxied by the share of Chinese imports in apparent consumption (Bernard et al.,

Year	All (%)	Gender	r (%)		Age (%)			Skill (%)		
		Female	Male	16-29	30-44	45-65	High	Medium	Low	
1997	7.2	7.6	7.1	9.5	6.1	7.2	7.2	6.7	7.3	
1998	7.0	8.1	6.7	8.4	6.3	7.0	6.0	7.4	7.1	
1999	7.9	10.1	7.4	7.7	7.4	8.8	7.6	8.5	7.9	
2000	6.1	6.6	5.9	5.1	5.5	7.4	6.6	6.6	5.8	
2001	7.1	8.0	6.8	6.3	6.2	8.8	6.9	7.3	7.1	
2002	7.4	8.9	7.0	7.0	6.8	8.6	7.0	7.8	7.4	
2003	7.2	8.7	6.7	6.7	6.5	8.4	6.1	7.5	7.3	
2004	7.5	9.0	7.0	7.6	6.8	8.4	8.0	7.6	7.3	
2005	7.8	8.9	7.4	7.3	6.9	9.4	8.8	7.7	7.7	
2006	7.6	9.5	7.0	7.4	7.1	8.4	7.3	7.8	7.6	
2007	7.5	8.8	7.0	7.7	7.2	7.8	6.6	8.0	7.5	
2008	10.1	11.6	9.6	12.4	9.4	9.9	7.5	9.1	10.9	
2009	12.6	13.9	12.1	14.8	11.5	13.1	8.8	12.0	13.5	
2010	9.5	10.1	9.3	10.0	8.9	10.1	8.2	9.9	9.6	
2011	9.1	9.7	8.9	11.8	8.2	9.4	7.3	9.3	9.4	
Total	8.1	9.4	7.7	8.1	7.5	8.9	7.3	8.2	8.2	

Table 2. Unconditional probabilities of becoming unemployed.

Source: authors' calculations based on MCVL.

2006). Analytically,

$$CP_{China,k} = \frac{M_{China,k}}{P_k - X_k + M_k}$$
(1)

where $M_{China,k}$ are Chinese imports in industry k, P_k is the Spanish production in industry k, X_k are Spanish exports in industry k, and M_k are total Spanish imports in industry k. Spanish exports and imports data come from the Agencia Tributaria Trade database and production data come from the Spanish Statistical Institute Industrial Survey database.

Figure 1 presents the evolution of Chinese imports' competitive pressure on the Spanish manufacturing sector during the period 1997-2011. As shown in the figure, there has been a very large increase in competitive pressure from Chinese imports. In the year 1997 the share of Chinese imports only represented 0.96% of Spanish manufacturing apparent consumption; in 15 years this share had multiplied by almost five, reaching the 4.70% of Spanish manufacturing apparent consump-

tion. If we only consider those industries where Chinese imports represented at least 1% of apparent consumption in 1997, Figure 2 shows that the competitive pressure of Chinese imports has increased in all industries. However, the increase in the competitive pressure from Chinese imports has been very different across industries. For example, in dressing and dyeing of fur (NACE 183) competitive pressure has increased in 32 percentage points, and in manufacture of office machinery and computers (NACE 300) in 31 percentage points. In contrast, in plastic products (NACE 252) and basic chemicals (NACE 241) the competitive pressure has increased only in 2 percentage points.

Our expectation is that workers occupied in industries that experienced a larger increase in the competitive pressure from Chinese imports had a larger increase in the probability of becoming unemployed than workers occupied in industries where the increase of competitive pressure from Chinese imports was lower. The next section will test the validity of this expectation.



Figure 1. Evolution of the share of Chinese imports in Spanish manufacturing apparent consumption, 1997-2011

Source: authors' calculation based on Agencia Tributaria, and INE Industrial Survey databases.



Figure 2. Share of Chinese imports in apparent consumption by industry: 2011 vs. 1997

Note: only includes industries where Chinese imports represented at least 1% of apparent consumption in 1997. Source: authors' calculation based on Agencia Tributaria, and INE Industrial Survey databases.

3. Econometric analysis

3.1 Imports from China and the probability of becoming unemployed

As our benchmark analysis we estimate the following econometric equation to determine the influence of Chinese import competition on the probability of becoming unemployed:

$$U_{it} = \beta_m China_{kt-1} + \beta'_i X_i + \beta'_c X_c + \beta'_e X_e + \beta_t$$
⁽²⁾

where U_{it} takes the value of 1 if the worker gets unemployed at year t, and $China_{kt-1}$ is the share of Chinese imports in the apparent demand for industry k at year t-1. This variable is lagged one year in order to avoid the effects of shocks that can simultaneously affect the probability of becoming unemployed and the share of Chinese imports in apparent demand; moreover, firms might decide to lay-off workers once they observed an increase in import competition from China.⁶

X_i is a vector of individual variables that previous studies have identified as contributing to the probability of becoming unemployed. First, we control for workers' age, creating three age categories: workers between 16 and 29 years; workers between 30 and 44 years; and workers between 45 and 65 years. The excluded category in the regression is workers between 16 and 29 years. Second, we control for the sex of the worker; we add a dummy variable that takes the value of 1 if the worker is male and zero if the worker is female. Third, we add a dummy variable to control for the nationality of the worker; the dummy variable takes the value of 1 if the worker does not have the Spanish nationality and zero otherwise. X_c is a vector of work-contract related variables that might influence the probability of becoming unemployed. It includes worker's occupation, classified into three different skilllevels. Workers performing jobs that require a university degree, high-level managers, and administrative and workshop bosses are included in the high-skill group; support staff without university degree, administrative officials and auxiliaries, subordinates and 1st

level officials are included in the medium-skill group; the rest of occupations are included in the low-skill group. This last skill-group is the category excluded in the regression. The vector also includes the length of the working day; the dummy variable takes the value of 1 if the worker is employed the whole working day and zero otherwise. Another dummy controls whether the permanent (open-ended) contract has any type of discount in terms of social security costs. The dummy variable takes the value of 1 if the permanent (open-ended) contract has a discount and zero otherwise. We also include the experience of the worker in the firm, measured by number of days since her contract begun. Finally, X is a vector of employer-related variables that might influence the probability of becoming unemployed.⁷ It includes the size of firms, proxied by the number of employees, that we classify in three categories: small firms (1-49 employees), medium-sized firms (50-499 employees) and large firms (500 or more employees). The excluded category in the regression is small firms.⁸ Estimations also include dummy variables for each year.

The econometric estimations are carried out pooling data for the period 1997-2011. In all regressions standard errors are clustered at 3-digit industries, the level at which we observe competition from China. We estimate the equation with a logistic model. Table 3 presents the results of the econometric estimations. First, we estimate the model without the control vectors. Afterwards, one by one, we include the observable controls related to individual, contract and employer characteristics. Column (1) shows that import competition from China is positive and statistically significant. This result confirms our expectation that workers occupied in manufacturing industries with a higher exposure to Chinese import competition have a larger probability of becoming unemployed than workers occupied in industries with a lower exposure to Chinese competition. Column (2) introduces the observable individual level characteristics. The coefficient for Chinese imports remains

⁶ We repeated all estimations using contemporaneous values and two-year lagged values of the share of Chinese imports in apparent demand. Results were not altered.

⁷ In MVCL firms are identified by the code assigned by the Spanish Social Security. If a firm has plants in different Spanish provinces, the Social Security provides a code for each province. Hence, in the case of multi-plant firms, the code identifies province-level plants. 8 Our first observation for the size of the firm is year 2004. For previous years we assume that firms do not alter their size status

positive and statistically significant. We observe that workers between 30-44 years have a lower probability of becoming unemployed; however, older workers have a higher probability of becoming unemployed. We also find that being male reduces the probability of becoming unemployed, whereas being foreigner increases it. In Column (3) we introduce the characteristics of the contract. The coefficient for Chinese imports remains positive and statistically significant. We find that performing tasks that require high-skills reduces the probcontract and employer-level characteristics. To quantify the contribution of Chinese import competition to the probability of becoming unemployed, we calculate the marginal effect of the share of Chinese imports at average values. If we multiply this marginal effect (0.001) by a standard deviation increase in import competition from China (7.4 percentage points), the probability of becoming unemployed rises in (0.001*7.4) 0.0074 percentage points. If we divide this figure by the unconditional probability of becoming unemployed (0.081), we obtain that a standard deviation increase in import competition from China raises the

 Table 3. Benchmark econometric estimations, 1997-2011

	(1)	(2)	(2)	(4)	(5)	(6)
	(1)	(2)	(3)	(4)	(3)	(0)
Imports China	0.0225*** (0.0051)	0.0198*** (0.0042)	0.0195*** (0.0039)	0.0148*** (0.0034)	0.0048*** (0.0013)	0.0338*** (0.0090)
Age 30-44		-0.1306*** (0.0199)	0.0087 (0.0178)	0.0256 (0.0170)	0.0031 (0.0023)	-0.1333** (0.0572)
Age 45-65		0.0887** (0.0365)	0.2977*** (0.0341)	0.3361*** (0.0384)	-0.0335***(0.0033)	-0.3779*** (0.0879)
Male		-0.1936*** (0.0027)	-0.1571*** (0.0298)	-0.1661*** (0.0237)		
Foreigner		0.3288*** (0.0465)	0.1676*** (0.0468)	0.0968*** (0.0285)		
High-skill occup.			-0.0586 (0.0379)	0.0502 (0.0313)	-0.0938*** (0.0088)	-7.1966*** (0.3055)
Medium-skill occup.			0.0068 (0.0254)	0.0068 (0.0189)	-0.0551*** (0.0062)	-2.9649*** (0.2316)
Whole working			-0.0836*** (0.0280)	0.1537*** (0.0262)	0.0104 (0.0104)	2.2077*** (0.2420)
Discount-contract			0.3655*** (0.0166)	0.2512*** (0.0185)	0.0041 (0.0050)	-1.7537*** (0.1493)
Experience			-0.0807*** (0.0072)	-0.0881*** (0.0073)	-0.0145*** (0.0012)	0.4668*** (0.0139)
Medium firm				-0.7120*** (0.0394)	-0.1748*** (0.0149)	-1.6000*** (0.0939)
Large firm				-1.0759*** (0.0728)	-0.2448*** (0.0204)	-4.6199** (0.2544)
Model	Logit	Logit	Logit	Logit	Linear Probability	Logit fixed-effects
Observations	1,031,275	1,031,275	1,031,275	1,012,035	1,012,035	377,441
R-squared	0.007	0.010	0.018	0.038	0.323	

Note: Imports China is lagged one year. All regressions include year-specific fixed effects. Regression in Column 5 includes workerspecific and firm-specific fixed effects. Regression in Column (6) includes worker-firm specific fixed-effects. Robust standard errors clustered by 3-digit industries in parentheses. ***, ** statistically significant at 1% and 5% respectively.

ability of becoming unemployed; however, it is not statistically significant. We also find that working the full working-day and commanding a larger experience in the firm reduce the probability of becoming unemployed; in contrast, having a contract with discounts in social security costs increases the probability of becoming unemployed. Finally, we include the characteristics of the employer. The coefficient for import competition from China remains positive and statistically significant. We can see that the probability of becoming unemployed is lower for workers occupied in medium and large manufacturing firms. We should note that the fit of the model is very low in all estimations.

Results reported in Table 3 show that import competition from China increases the probability of becoming unemployed. The negative effect of Chinese competition remains once we control for observable individual, probability of becoming unemployed by 9% (0.0074/0.081).

The econometric estimations reported in Column (4) control for observed individual, contract and employer characteristics. However, there might be unobserved worker-level characteristics, such as the inner talent of the worker, or unobserved firm-level characteristics, such as firm's innovative capacity, that might also affect the probability of becom-ing unemployed. In order to identify more precisely the contribution of Chinese import competition on the probability of becoming unemployed, in Column (5) we estimate the model with worker-specific and firm-specific fixedeffects. As explained in Greene (2008) and in Cameron and Trivedi (2009), in short panels with a large number of dummy vari-ables, non-linear models, such as logit, might lead to biased and inconsistent estimators due to an incidental parameters problem. Hence, we use a linear probability model to perform

this estimation.⁹ Due to the large number of dummy variables, we estimate the model using the high-dimensional fixed-effects routine proposed by Guimarães and Portugal (2010).¹⁰ Worker-level fixed effects absorb the workerrelated variables that do not change during the period of analysis, such as sex and nationality. As shown in Table 3 -Column (4), the coefficient Imports China remains positive and statistically significant, once we control for worker-specific and firm-specific fixed effects. Results also show that workers in the 45-65 age-cohort have a lower probability of becoming unemployed. Performing tasks that require medium and high-skills, working in medium of large firms, and having a longer experience also lead to a lower probability of becoming unemployed. According to the coefficient reported in Column (5), a standard deviation increase in Chinese import competition raises the probability of becoming unemployed, relative to the unconditional probability, by 44% ((0.0048*7.4)/0.081)).

In order to improve the control of nonobservable factors, in Column (6) we present the results of estimating the equation with a fixed-effects logit model. The panel estimation controls for firm-worker specific fixed-effects. The limitation of this analysis is that the sample is narrowed down to the workers that have moved to unemployment; workers that keep their work-contract with the same firm alive during the period of analysis do not experience any variation in their dependent variable and are dropped from the estimation. In particular, the sample is reduced from 1,012,035 observations to 377,441 observations. As shown in Column (6), the coefficient Imports China is positive and statistically significant, denoting that the probability of transitioning to unemployment is positively linked with the rise in competition from China.

To further analyze the effect of imports from China on the probability of becoming unemployed, we replicate the estimations on Table 3 separately for different skill-levels and age-groups. The results presented in Panel A of Table 4 confirm that workers in skill-intensive occupations have a lower probability of becoming unemployed due to import competition from China. The coefficient on imports from China is not significant at conventional levels for high-skill occupations, but is positive and statistically significant for medium-skill and low-skill occupations. Moreover, the coefficient becomes larger as the skill-content of the occupation is reduced. As shown by previous studies, firms can respond to import competition from China outsourcing labor-intensive tasks to low-wage countries and raising products' technological content (Bloom et al., 2012; Ebenstein et al., 2012a). These strategies reduce the demand for unskilled occupations and raise the demand for more skill-intensive occupations, explaining the lower impact of Chinese competition on occupations demand-ing more skills. The marginal effect of the share of Chinese imports at average values for medi-um-skill and low-skill occupations is 0.0007 and 0.0010 respectively. Thus, the logit model implies that a standard deviation increase in import competition from China would raise the probability of becoming unemployed by 6.3% and 9% in a medium-skill and low-skill occupation respectively. When we estimate the model with worker-specific and firmspecific fixed effects (Panel B, Table 4) the results are rather similar. The coefficient for imports from China for the high-skill occupations is now significant at the 1% level and, again, the co-efficient becomes larger as the skill-content of the occupation is reduced. The coefficient es-timates in the linear probability model imply that a standard deviation increase in import competition from China would raise the prob-ability of becoming unemployed by 28.4% in a highskill occupation, by 32.5% in a medium-skill occupation and by 49.6% in a low-skill occupation. In contrast, when we estimate the model with worker-firm specific fixed effects (Panel C, Table 4), the estimated coefficient of imports from China is larger for the high-skill occupations and becomes lower as the skillcontent of the occupation is reduced. Nevertheless, this result should be interpreted with

⁹ An advantage of the linear probability model is that the estimated coefficients are easy to interpret, as they show how the probability of the event (in our case, transitioning from employment to unemployment) changes when the independent variable increases by one unit if the variable is continuous, or changes its status if the variable is dichotomic. However, the limitation of the linear probability model is that the effect of independent variables on the dependent variable is constant. In addition to that, the linear probability model can yield predicted probabilities below zero and above one. Moreover, the linear probability model is inherently heteroskedastic. In order to control for heteroskedasticity we estimate the model with clustered standard errors.

¹⁰ We want to thank Paulo Guimarães for his help in the application of this routine to our database.

caution since the reported estimates may be influenced by the previously mentioned sample selection bias, where workers that keep their work-contract with one firm alive during the period of analysis are dropped from the sample.

Attending to the estimation results by age group, the coefficient on imports from China is slightly higher for workers in the younger (age 16-29) and older (age 45-65) group and lower for workers in the middle group (age 30-44). For the logit model in Panel A (Table 4), the marginal effect of the share of Chinese imthe 16-29 age-cohort, by 48.3% for workers in the 30-44 age-cohort and by 48.2% for workers in the 45-65 age-cohort. When estimating the model with worker+firm fixed effects import competition from China is only significant for young workers.

Overall, results confirm that import competition from China is positively associated with a higher probability of becoming unemployed, even when we control for (timeinvariant) heterogeneity both at firm and worker level. Moreover, the impact of competition from China is substantial, increasing the

		By Skill level			By age group			
	High-Skill	Medium-Skill	Low-Skill	Age 16-29	Age 30-44	Age 45-65		
	(1)	(2)	(3)	(4)	(5)	(6)		
			PANEL	A: Logit				
Imports China	0.0062 (0.0058)	0.0109*** (0.0033)	0.0160*** (0.0031)	0.0168*** (0.0036)	0.0153*** (0.0037)	0.0163*** (0.0042)		
Observations R-squared	134,423 0.002	207,318 0.005	689,534 0.010	205,937 0.013	466,001 0.008	359,337 0.006		
	PANEL B: Linear Probability (worker-specific and firm-specific fixed effects)							
Imports China	0.0028*** (0.0009)	0.0036*** (0.0011)	0.0055*** (0.0015)	0.0070** (0.0037)	0.0049*** (0.0016)	0.0058*** (0.0012)		
Observations R-squared	132,019 0.35	202,740 0.34	677,276 0.32	202,363 0.44	457,877 0.39	351,795 0.35		
	PANEL C: Logit (worker-firm specific fixed effects)							
Imports China	0.0620** (0.0305)	0.0422* (0.0242)	0.0367*** (0.0112)	0.0926*** (0.0271)	0.0218 (0.0151)	0.0077 (0.0198)		
Observations	43,705	/4,262	257,034	48,072	122,291	0.0198		

Table 4. Econometric estimations by skill-level and age-group. 19	997-2011
---	----------

Note: Imports China is lagged one year. All regressions include worker, contract and employer characteristics as explanatory variables and specific fixed effects. Robust standard errors clustered by 3-digit industries in parentheses. ***, ** statistically significant at 1% and 5% respectively.

ports at average values is 0.0009, 0.0008 and 0.0012 for the group of age 16-29, 30-44 and 45-65 respectively. These values imply that a standard deviation increase in import competition from China would raise the probability of becoming unemployed by 8.2% for workers with age 16-29, by 7.9% for workers with age 30-44 and by 10% for workers with age 45-65. The reported estimates when we estimate the model with worker-specific and firm-specific fixed-effects (Panel B, Table 4) imply that a standard deviation increase in import competition from China would raise the probability of becoming unemployed by 64% for workers in

probability of becoming unemployed, relative to the unconditional probability, between 9% and 44%. Our result are in line with the conclusions reached by recent studies that find a strong negative relationship between import competition from China and the decline of manufacturing employment in the US (Autor et al. 2013, Ebenstein et al. 2012b, and Pierce and Schott, 2012), and in Spain (Donoso et al., 2014); and with studies that find a negative relationship between imports from low-wage countries and employment growth at the firmlevel in high-wage countries (Bernard et al., 2006; Mion and Zhu, 2013).

3.2 Technological progress, the financial crisis and alternative measures of trade exposure.

To analyze the robustness of our results, we perform some sensitivity analyses. In all sensitivity analyses we report the results of estimating a linear probabilistic model with worker-specific and firm-specific fixed effects, and a fixed-effects logit model. First, we analyze whether the negative effect of import competition from China is robust to the introduction of variables that capture technological progress in the Spanish manufacturing industries that may also increase the probability of becoming unemployed. In particular, we control for capital/labor ratio, total factor productivity, and share of information and communiables are lagged one year. As shown in Column (1) and (2) of Table 5, the coefficient for import competition from China remains positive and statistically significant. The coefficients for the technology variables are statistically not significant in the linear probabilistic model estimation; however, in the fixed-effects logit estimation we find that workers employed in firms that increase total factor productivity have a lower probability of becoming unemployed.

Second, in order to confirm that our results are not driven by the severe negative impact that the financial crisis had on the demand for labor in the Spanish manufacturing sector, we estimated the model for the period 1997-2007. As shown in Column (3) and (4) of Table 5 the

Table 5. Sensitivity analyses: technological progress and time-period avoiding the financial crisis

	Technologi	cal progress	Period 1997-2007		
	(1)	(2)	(3)	(4)	
Imports China	0.0094*** (0.0013)	0.1062*** (0.0201)	0.0067*** (0.0013)	0.0632*** (0.0175)	
Age 30-44	0.0028 (0.0025)	-0.1719*** (0.0637)	0.0017 (0.0025)	-0.3004*** (0.0701)	
Age 45-65	-0.0334*** (0.0037)	-0.3806*** (0.1008)	-0.0296*** (0.0035)	-0.3912*** (0.1159)	
High-skill occup.	-0.0979*** (0.0134)	-7.7907*** (0.3319)	-0.0912*** (0.0079)	-8.2030*** (0.3570)	
Medium-skill occup.	-0.0545*** (0.0083)	-3.6055*** (0.2539)	-0.0493*** (0.0060)	-3.5304*** (0.2682)	
Whole working day	0.0049 (0.0138)	1.3072*** (0.3364)	0.0015 (0.0104)	0.5221 (0.3916)	
Discount-contract	0.0060 (0.0063)	-2.6324*** (0.1757)	0.0100 (0.0048)	-2.7393*** (0.1856)	
Experience	-0.0135*** (0.0016)	0.3853*** (0.0151)	-0.0105*** (0.0011)	0.3314*** (0.0161)	
Medium firm	-0.1770*** (0.0223)	-1.7103*** (0.1222)	-0.1617*** (0.0224)	-1.6292*** (0.1781)	
Large firm	-0.2490*** (0.0323)	-4.0895*** (0.2956)	-0.2364*** (0.0383)	-2.9289*** (0.3611)	
K/L ratio	-0.0314 (0.0450)	0.2579 (0.8145)			
TFP	-0.0877 (0.0580)	-2.4533*** (0.4808)			
ICT	0.0206 (0.0176)	0.2886 (0.2810)			
Imports EU15					
-					
Model	Linear probabilistic	Fixed-effects logit	Linear probabilistic	Fixed-effects logit	
Observations	879,444	293,019	750,181	203,811	
R-squared	0.343		0.368		

Note: All regressions include year-specific fixed effects. Linear probabilistic regressions include worker-specific and firm-specific fixed effects. Imports China and technology variables lagged one year. In Columns (1) and (2) Imports China is calculated at the 2-digit industry level and the period of analysis is 1997-2009. In Columns (3) and (4) the period of analysis is 1997-2007. Robust standard errors in parentheses, clustered by 2-digit industries in Columns (1) and (2), and clustered by 3-digit industries in Columns (3) and (4). ***, ** statistically significant at 1% and 5% respectively.

cation technologies (ICT) in capital services.¹¹ As these data are at the 2-digit industry level, we recalculate the share of Chinese imports in apparent consumption for this aggregation level. Technological data are only available until 2009, so the period of analysis is restricted to 1997-2009. Trade and technological vari-11 TFP and ICT data are obtained from the Euklems database (www.euklems.net); capital/labor ratio data are obtained from the OECD STAN database (stats.oecd.org)

coefficient Imports China remains positive and statistically significant at the 1% level.

Lastly, we analyze the robustness of our previous results for different measures of trade exposure. First, we analyze whether import competition from China has, in fact, a higher negative impact on the demand for labor than import competition from other countries. To test this hypothesis we calculate the share of European Union 15 countries, except Spain, on the apparent consumption of Spanish manufacturing. As shown in Column (1) of Table 6, in the linear probability model, the coefficient import coefficient from China is almost four times larger than the coefficient import competition from European Union 15 countries. If we narrow down the sample to workers that transition to unemployment (Column (2) of Table 6), we also observe that this transition is more likely when there is a rise in import competition from China than when there is a rise in import competition from European Union 15 countries. These results confirm that import competition from China, being concentrated in labor-intensive products, has a much larger negative impact on the demand for labor than imports from high-wage countries. Second, to allow for the possibility that imports from China are compensated by Spanish exports to China we calculate a net imports from China as a share of apparent consumption indicator, subtracting Spanish exports to China from Spanish imports from China in the numerator of equation 1. Since Spanish imports from China are much larger than Spanish exports to China we do not expect a great variation in our previous findings. The results for the impact of net imports from China on the probability of becoming unemployed are reported in Column (3) and (4) of Table 6. It can be appreciated that the coefficient of net imports from China is positive and statistically significant at the 1% level. The coefficient in Column (3) is about 23 percent smaller and similarly precisely estimated than the coefficient estimated with a model that uses gross imports rather than net imports (Column 5 - Table 3). Its magnitude implies that a standard deviation increase in net imports from China would raise the probability of becoming unemployed by 42%. Similarly, the coefficient of net imports from China when including worker-firm specific fixed effects (Column 4) is a 46% smaller than its homologous in Column (6) of Table 3. Third, our initial measure of import competition from China includes both final goods and intermediate goods. If imports from China increase the variety of inputs that can be used by Spanish firms, their productivity may increase along with their demand for labor. To focus on the effect of import competition in final goods,

we replace the share of imports from China over apparent consumption by the share of total imports less imports of intermediate inputs over apparent consumption. Imported intermediate goods by industry were obtained by combining trade data at the 2-digit level with the Spanish input-output table database. In Column (5), the coefficient on imports from China is 0.0066, 38% higher than in Column (5) of Table 3 and similarly precisely estimated. Moreover, the estimated coefficient implies that a standard deviation increase in imports from China net of intermediate inputs would raise the probability of becoming unemployed by almost 46%.

Equation 1 might have some limitations since import competition from China might show up through other labor market outcomes. For example, if wages adjust downwards reducing costs in the manufacturing sector, there would be a lower increase in imports from China, since part of this competitive pressure would be absorbed by the fall in wages. However, as analyzed in the next section, manufacturing wages do not seem to be influenced by import competition. This result indicates that during the period of analysis manufacturing wages were rigid, at least downwards, and demand shocks were absorbed mainly through quantities, validating the use of equation 1.

4. The impact of import competition from China on wages and sectoral real-location

In this section we analyze the impact of import competition from China on other labor market variables such as wages and sectoral reallocation. To analyze the impact on wages we draw data from the fiscal dataset attached to the MCVL main dataset. The fiscal dataset reports the total annual taxable labor income per worker for a particular year. The limitation of this dataset is that it only provides data for the period 2004-2011. In addition to that, data is not available for all workers. Following Bonhomme and Hospido (2012), we use daily wage as our wage measure, calculated as the ratio between annual taxable income and the number of days worked in a particular year.

Table 7 displays the results of the regression analyses. In Column (1) we introduce

	Imports from	m the EU15	Net imports fr	om China	Final goods imports from China	
	(1)	(2)	(3)	(4)	(5)	(6)
Imports China	0.0046***	0.0337***	0.0037***	0.0183***	0.0066***	0.0353***
-	(0.0012)	(0.0089)	(0.0014)	(0.0062)	(0.0013)	(0.0127)
Age 30-44	0.0032	-0.1313**	0.0030	-0.1359**	0.0032	-0.1353**
	(0.0023)	(0.0573)	(0.0023)	(0.0572)	(0.0023)	(0.0572)
Age 45-65	-0.0335***	-0.3742***	-0.0335***	-0.3771***	-0.0335***	-0.3763***
	(0.0032)	(0.0979)	(0.0033)	(0.0879)	(0.0033)	(0.0879)
High-skill occup.	-0.0937***	-7.1870***	-0.0937***	-7.2012***	-0.0940***	-7.1993***
	(0.0088)	(0.3053)	(0.0088)	(0.3056)	(0.0089)	(0.3056)
Medium-skill occup.	-0.0551***	-2.966***	-0.0548***	-2.9598***	-0.0550***	-2.9593***
	(0.0062)	(0.2315)	(0.0061)	(0.2315)	(0.0063)	(0.2316)
Whole working day	0.0105	2.2170***	0.0103	2.2043***	0.0105	2.2072***
	(0.0104)	(0.2420)	(0.0104)	(0.2420)	(0.0104)	(0.2421)
Discount-contract	0.0041	-1.7459***	0.0042	-1.7571***	0.0039	-1.7568***
	(0.0050)	(0.1487)	(0.0050)	(0.1493)	(0.0050)	(0.1493)
Experience	-0.0146***	0.4661***	-0.0146***	0.4670***	-0.0145***	0.4671***
	(0.0012)	(0.0138)	(0.0012)	(0.0138)	(0.0012)	(0.0138)
Medium firm	-0.1748***	-1.6129***	-0.1752***	-1.6010***	-0.1753***	-1.6020***
	(0.0149)	(0.0941)	(0.0149)	(0.0939)	(0.0150)	(0.0939)
Large firm	-0.2466***	-4.6171***	-0.2458***	-4.6201***	-0.2454***	-4.6241***
	(0.0206)	(0.2534)	(0.0205)	(0.2543)	(0.0203)	(0.2545)
Imports EU15	0.0012**	0.0268**				
	(0.0005)	(0.0039)				
Model	Linear	Fixed-effects	Linear	Fixed-effects	Linear	Fixed-effects
	probabilistic	logit	probabilistic	logit	probabilistic	logit
Observations	1,012,035	377,441	1,012,035	377,465	1,012,035	377,465
R-squared	0.32		0.32		0.32	

Table 6. Sensitivity analyses: alternative measures of trade exposure, 1997-2011

Note: All regressions include year-specific fixed effects. Linear probabilistic regressions also include worker-specific and firm-specific fixed effects. Imports China lagged one year. Robust standard errors in parentheses, clustered by 3-digit industries. ***, ** statistically significant at 1% and 5% respectively.

	(1)	(2)	(3)	(4)	(5)	(6)
Imports China	-	-0.0085**	-0.0082***	-0.0063***	0.0010	0.0011
-	0.0100**	(0.0036)	(0.0025)	(0.0020)	(0.0097)	(0.0010)
	(0.0039)					
Age 30-44		0.2370***	0.1268***	0.1109***	0.0358**	0.0364***
		(0.0146)	(0.0095)	(0.0093)	(0.0042)	(0.0042)
Age 45-65		0.3659***	0.2307***	0.2066***	0.0234***	0.0237***
		(0.0218)	(0.0157)	(0.0138)	(0.0053)	(0.0054)
Male		0.2012***	0.2441***	0.2494***		
		(0.0353)	(0.0205)	(0.0169)		
Foreigner		-0.2161***	-0.0893***	-0.0544***		
-		(0.0043)	(0.0043)	(0.0191)		
High-skill occup.			0.7160***	0.6551***	0.0797***	0.1301**
			(0.0245)	(0.0207)	(0.0299)	(0.0646)
Medium-skill occup.			0.2954***	0.2848***	0.0550**	0.1136**
_			(0.0274)	(0.0261)	(0.0228)	(0.0493)
Discount-contract			-0.1431***	-0.1013***	-0.0131	0.0030
			(0.0132)	(0.0115)	(0.0136)	(0.0210)
Experience			0.0219***	0.0262***	0.0385***	0.0538***
-			(0.0019)	(0.0017)	(0.0038)	(0.0043)
Medium firm				0.2417***	0.0253***	0.0248***
				(0.0162)	(0.0083)	(0.0083)
Large firm				0.4137***	0.0288**	0.0266*
				(0.0225)	(0.0142)	(0.0147)
Observations	344,194	344,194	344,194	344,035	344,035	343,682
R-squared	0.02	0.07	0.23	0.28	0.93	0.93

Table 7. Impact of import competition from China on wages, 2004-2011

Note: The dependent variable is daily wage in logs. Imports China is lagged one year. All regressions include year-specific fixed effects. Column (5) also includes worker-specific and firm-specific fixed effects. Column (6) includes worker+firm-specific fixed effects. Robust standard errors clustered by 3-digit industries in parentheses. ***, **, * statistically significant at 1%, 5% and 10% respectively.

imports from China and year dummies as the only independent variables. We can see that import competition from China is negatively associated with the wage level. In Column (2) we introduce workers' personal characteristics, such as age, sex and nationality. Import competition from China remains negative and statistically significant. We observe that workers in the 30-44 cohort and workers in the 45-65 cohort command higher wages than young workers. Males have a higher daily wage than females and foreigners have a lower daily wage than workers with the Spanish nationality. In Column (3) we introduce the characteristics of the contract. Import competition from China remains negative. We can see that workers performing high-skill and mediumskill tasks have a higher daily wage than workers performing low-skill tasks. In addition to that, workers with contracts with social security cost deductions have a lower daily wage; in contrast, a larger experience is correlated with a higher daily wage. In Column (4) we control for employer-level characteristics. We can see that import competition from China remains negative and statistically significant. The results show that workers employed in medium and large firms command a higher wage than workers employed in small firms. In Column (5) we control for worker-specific and firm-specific fixed effects. Import competition from China becomes statistically not significant. This result shows that once we control for worker and firm-specific unobservable effects there is no longer a negative association between wages and import competition from China. This result is confirmed in Column (6), when we control for worker-firm specific fixed effects.

In order to analyze the impact of imports from China on sectoral reallocation, we clas-



Figure 3. Imports from China and switching from manufacturing industries

Note: only includes industries where the mean Chinese imports as percentage of apparent consumption along the period 1997-2011 was at least 1%.

Source: authors' calculation based on MCVL.

sify workers in our sample into two distinct categories, for each year between 1997 and 2011. The first category is workers that switch to an employment outside manufacturing; this category also includes manufacturing workers that become unemployed and switch to an employment outside manufacturing if they were unemployed less than one year. The second category is workers who remain employed in the manufacturing sector. Following the same approach as in section 3.1, this classification allows us to explore the impact of import competition from China on the probability of switching to an employment outside manufacturing. Our sample consists of a total of 907,861 observations of which a 4.4% correspond to workers switching outside the manufacturing sector. The manufacturing indusclothing apparel and accessories (3.6%). For illustration purposes, Figure 3 shows a scatter plot between the number of switching spells by industry as percentage of total switching spells and the mean lagged imports from China as percentage of apparent consumption along the 1997-2011 period. The figure does not show a clear positive relationship between switching spells and import competition from China. Among the destination industries, other entrepreneurial activities received 14.3% of total switchers from manufacturing, while 13% and 11% of workers switched out from manufacturing to an employment in construction and wholesale trade and trade intermediaries respectively.

	(1)	(2)	(3)	(4)	(5)	(6)
Imports China	0.0119**	0.0105*	0.0109**	-0.0006	0.0001	0.0280**
1	(0.0063)	(0.0060)	(0.0052)	(0.0051)	(0.0005)	(0.0125)
Age 30-44	× ,	-0.5650***	-0.2371***	-0.2562***	0.0278***	0.3921***
- C		(0.0361)	(0.0324)	(0.0353)	(0.0020)	(0.0742)
Age 45-65		-1.2630***	-0.7940***	-0.9432***	0.0144***	-0.1701
		(0.0360)	(0.0363)	(0.0421)	(0.0021)	(0.1303)
Male		-0.0502	0.0771*	0.0838*		
		(0.0535)	(0.0437)	(0.0522)		
Foreigner		0.6973***	0.3546***	0.2949***		
		(0.0693)	(0.0661)	(0.0520)		
High-skill occup.			0.0659***	0.1994***		-7.3322***
			(0.0662)	(0.0762)		(0.5378)
Medium-skill occup.			0.1658***	0.1991***		-2.3929***
· · ·			(0.0439)	(0.0483)		(0.3862)
Whole working			-0.3517***	-0.2110***	0.0028	0.0270
			(0.0602)	(0.0661)	(0.0151)	(0.3626)
Discount-contract			0.7654***	0.7655***	0.0131	1.4563***
			(0.0295)	(0.0258)	(0.0091)	(0.1576)
Experience			-0.2124***	-0.2389***	-0.0102**	4.4370***
-			(0.0110)	(0.0154)	(0.0045)	(0.0654)
Medium firm				-0.4181***	-0.0401	0.3919
				(0.0402)	(0.0906)	(0.3927)
Large firm				-0.8068***	-0.0551	1.2876
-				(0.1361)	(0.0660)	(0.9988)
Model	Logit	Logit	Logit	Logit	Linear	Logit fixed-
	_	-	-	-	Probability	effects
Observations	907,861	907,861	907,861	730,560	730,560	93,415
R-squared	0.007	0.035	0.067	0.086	0.40	

Note: Imports China is lagged one year. All regressions include year-specific fixed effects. Regression in Column 5 includes worker-specific and firm-specific fixed effects. Robust standard errors clustered by 3-digit industries in parentheses. ***, ** statistically significant at 1% and 5% respectively.

tries recording a higher number of switches are rubber products (4.9% of total switches), furniture (4.4%), leather clothes (3.8%) and

The results of the regression analyses are reported in Table 8. In Columns (1) to (4) we report the results from the logistic model for different sets of explanatory variables. When we estimate the model with no controls except year dummies, the coefficient on imports from China is positive and significant at the 5% level, indicating that import competition increases the probability of switching to an employment outside manufacturing. In Column (2) and Column (3) we introduce workers' personal characteristics and the characteristics of the contract respectively. The coefficient on imports from China remains positive and statistically significant. Moreover, the probability of switching to an employment outside manufacturing reduces with age and increases as the skill component of the occupation becomes lower. Also, those workers with larger experience have a lower probability of switching. However, when we control for the size of the firm (Column 4), the coefficient of imports from China is not significant at conventional levels. This result holds when we estimate the model with worker-specific and firm specific fixed-effects (Column 5) by means of a linear probability model. In Column (6) we present the results for the fixed-effects logit model. The parameter estimate of imports from China is now significant at the 5% level and positive.

Overall, the presented results show that the evidence of a positive relationship between increasing import competition from China and the probability of switching to an employment outside manufacturing is rather weak. Thus reallocation of workers from manufacturing to other sectors along the 1997-2011 period was plausibly due to causes other than increasing import competition from China like wage increase differentials, as was probably the case for those workers switching to an employment in the construction sector.

5. Conclusions and policy implications

During the period 1997-2011 import competition from China in the Spanish manufacturing sector multiplied by five. In this paper, we analyzed whether this severe increase in import competition from China raised the probability of becoming unemployed among Spanish manufacturing workers. To test this hypothesis we use a dataset that comprises the working lives of 141,000 manufacturing workers. This dataset allows us to control for heterogeneity across workers and firms, enabling a better identification of the effect of Chinese import competition on the probability of becoming unemployed.

Our estimations show that there is a strong positive association between a higher import competition from China and the probability of becoming unemployed. We estimate that a standard deviation increase in import competition from China raises the probability of becoming unemployed between 0.8 and 3.5 percentage points, which represents between a 9% and a 44% increase over the unconditional probability of becoming unemployed. We do not identify any effect of import competition from China on wages. Also, our estimations show that there is weak evidence of a positive association between a higher import competition from China and the probability of switching to an employment outside the manufacturing sector.

As explained by Wood (2004) there are four different interventions that might be used to address the reduction in demand for manufacturing workers, and especially for unskilled workers, due to the increase in import competition from China. The first intervention is to raise barriers to trade against imported products from China. Although this intervention has popular appeal, it is not recommended for two reasons. On the one hand, the obligations that Spain has assumed as a member of the World Trade Organization preclude the unilateral introduction of higher barriers to trade. On the other hand, higher barriers to trade would limit the benefits from international trade, which recent studies show to be sizable (Feyrer 2009a and 2009b). Other policy interventions have a lower negative impact on efficiency and better target the collective of workers that are negatively affected by import competition from China. The second and more promising policy intervention would be to reduce the number of unskilled workers. From a wider perspective, this can be achieved rising the investment in education and focusing in the development of those skills and competences that are difficult to offshore or automate (Blinder, 2009; Autor and Dorn, 2013). From a narrower perspective, the reduction in the number of unskilled workers can be achieved upgrading their skills

through training. However, this policy has limitations, since it is a long-term measures and it is not always feasible to transform an unskilled worker into a skilled worker. This limitation calls for the use of other active labor market policies, such as job-search assistance, that can be implemented in the short-run (Jansen and Lee, 2007). The last policy intervention is to support the income of the unemployed. This policy is applied in Spain and in the majority of European Union countries. However, this policy has some limitations as well. If not properly designed it might discourage the return to work.

References

- Autor, D.H. and Dorn, D. (2013). The Growth of Low-Skill Service Jobs and the Polarization of the US Labor Market. American Economic Review, 103 (5), 1553-1597.
- Autor, D.H., Dorn, D. & Hanson, G.H. (2013). The China Syndrome: Local Labor Market Effects of Import Competition in the United States. American Economic Review, 103 (6), 2121-2168.
- Bachmann, R. & Braun, S. (2011). The impact of international outsourcing on labour market dynamics in Germany. Scottish Journal of Political Economy, 58 (1), 1-28.
- Berman, E., Bound, J. & Machin, S. (1998). Implications of Skill-Biased Technological Change: International Evidence. Quarterly Journal of Economics, 113 (4), 1245-1280.
- Bernard, A.B., Jensen, J.B. & Schott, P.K. (2006). Survival of the best fit: Exposure to low-wage countries and the (uneven) growth of U.S. manufacturing plants. Journal of International Economics, 68 (1), 219-237.
- Blinder, A.S. (2009). How Many US Jobs Might be Offshorable? World Economics, 10 (2), 41-78.
- Bloom, N., Draca, M., and Van Reenen, J. (2011). Trade induced technical change? The impact of Chinese imports on innovation, IT and productiviy. NBER working paper No. 16717, National Bureau of Economic Research, Cambridge, MA.
- Bonhomme, S. & Hospido, L. (2013). Earnings Inequality in Spain: New Evidence Using Tax Data. Applied Economics, 45 (30), 4212-4225.
- Cameron, A.C. & Trivedi, P.K. (2009). Microeconomics using Stata. College Station, Texas: Stata Press.
- Díez-Catalán, L., and Villanueva, E. (2012). Collective bargaining and unemployment during the great recession: Evidence for Spain. Available at http://www.sole-jole.org/13324.pdf.
- Donoso, V., Martín, V. y Minondo, A. (2014). "Do Differences in the Exposure to Chinese Imports Lead to Differences in Local Labour Market Outcomes? An Analysis for Spanish Provinces, Regional Studies, DOI: 10.1080/00343404.2013.879982
- Ebenstein, A., Harrison, A., McMillan, M. & Phillips, S. (2012). Estimating the Impact of Trade and Offshoring on American Workers Using Current Population Surveys. Review of Economics and Statistics, forthcoming.
- Egger, P., Pfaffermayr, M. & Weber, A. (2007). Sectoral adjustments of employment to shifts in outsourcing and trade: Evidence from a dynamic fixed effects multinomial logit model. Journal of Applied Econometrics, 22 (3), 559-580.
- Feenstra, R.C. & Hanson, G.H. (1996). Globalization, Outsourcing, and Wage Inequality. American Economic Review, 86 (2), 240-245.
- Feenstra, R.C. & Hanson, G.H. (1999). The Impact of Outsourcing and High-Technology Capital on Wages: Estimates for the U.S., 1972-1990. Quarterly Journal of Economics, 114 (3), 907-940
- Feyrer, J. (2009a). "Trade and Income. Exploiting Time Series in Geography". NBER Working Paper 14910, National Bureau of Economic Research.
- Feyrer, J. (2009b). "Distance, Trade, and Income The 1967 to 1975 Closing of the Suez Canal as a Natural Experiment". NBER Working Paper 15557, National Bureau of Economic Research.
- Geishecker, I. & Görg, H. (2008). Winners and losers: a micro-level analysis of international outsourcing and wages. Canadian Journal of Economics, 41 (1), 243-270.
- Görg, H. & Görlich, D. (2012). Trade and labour market outcomes in Germany. In Lippoldt, D. (Ed.), Policy Priorities for International Trade and Jobs (pp. 193-236). Paris: OECD.
- Green, W.H. (2008). Econometric Analysis. 6th Edition. New Jersey: Pearson.
- Guimarães, P. & Portugal, P. (2010). A simple feasible procedure to fit models with high-dimensional fixed effects. Stata Journal, 10 (4), 628-649.
- Harrison, A. & McMillan, M. (2011). Offshoring Jobs? Multinationals and U.S. Manufacturing Employment. The Review of Economics and Statistics, 93 (3), 857-875.
- Hummels, D., Jorgenssen, R., Munch, J. & Xiang, C. (2011). The Wage and Employment Effects of Outsourcing: Evidence from Danish Matched Worker-Firm Data. NBER Working Paper 17496, National Bureau of Economic Research, Cambridge, MA.
- Iacovone, L., Rauch, F., and Winters, L. A. (2013). Trade as engine of creative destruction: Mexican experience with Chinese competition. Journal of International Economics, 89(2), 379–392.

- Jansen, M. and Lee, E. (2007). Trade and Employment. Challenges for Policy Research. Geneva: World Trade Organization and International Labour Office.
- Krugman, P.R. (2008). Trade and Wages, Reconsidered. Brookings Papers on Economic Activity, 2, 103-137.
- Krugman, P.R. & Lawrence, R.Z. (1994). Trade, Jobs and Wages. Scientific American, April, 22-27.
- Leamer, E.E. (1998). In search of Stolper-Samuleson linkages between international trade and lower wages. In Collins, S. (Ed.), Imports, Exports and the American Worker (pp. 141-206). Washington DC: Brookings Institution Press.
- Lurweg, M. & Uhde, N. (2010). International Trade and Individual Market Perspectives. A Micro-Level Analysis of German Manufacturing Workers. CAWM Discussion Paper No. 31.
- Medina, E., Herrarte, A. & Vicéns, J. (2010). Inmigración y desempleo en España: Impacto de la crisis económica. Información Comercial Española. Revista de Economía, 854, 37-48.
- Mion, G. & Zhu, L. (2013). Import competition from and offshoring to China: A curse or blessing for firms?. Journal of International Economics, 89 (1),202-215.
- Montero, J.M. & Regil, A. (2012). Análisis de la evolución del desempleo desde una perspectiva de flujos laborales. Boletín Ecónomico del Banco de España, Marzo, 33-43.
- Pierce, J. R. and & Schott, P. K. (2012). The surprisingly swift decline of U.S. manufacturing employment. NBER working paper No. 18655, National Bureau of Economic Research, Cambridge, MA.
- Wood, A. (1994). North-South Trade, Employment and Inequality. Changing Fortunes in a Skill Driven World. IDS Development Studies Series, Oxford: Clarendon Press.
- Wood, A. (1995). How Trade Hurt Unskilled Workers. Journal of Economic Perspectives, 9 (3), 57-80.

Últimos títulos publicados

DOCUMENTOS DE TRABAJO "EL VALOR ECONÓMICO DEL ESPAÑOL"

- DT 16/11 Fernández Vítores, David: El papel del español en las relaciones y foros internacionales: Los casos de la Unión Europea y las Naciones Unidas. DT 15/11 Rupérez Javier: El Español en las Relaciones Internacionales. DT 14/10 Antonio Alonso, José; Gutiérrez, Rodolfo: Lengua y emigración: España y el español en las migraciones internacionales. DT 13/08 de Diego Álvarez, Dorotea; Rodrigues-Silveira, Rodrigo; Carrera Troyano Miguel: Estrategias para el Desarrollo del Cluster de Enseñanza de Español en Salamanca. DT 12/08 Quirós Romero, Cipriano: Lengua e internacionalización: El papel de la lengua en la internacionalización de las operadoras de telecomunicaciones. DT 11/08 Girón, Francisco Javier; Cañada, Agustín: La contribución de la lengua española al PIB y al empleo:
- una aproximación macroeconómica.
- **DT 10/08** Jiménez, Juan Carlos; Narbona, Aranzazu: *El español en el comercio internacional.*
- **DT 09/07** Carrera, Miguel; Ogonowski, Michał: *El valor económico del español: España ante el espejo de Polonia.*
- **DT 08/07** Rojo, Guillermo: *El español en la red.*
- **DT 07/07** Carrera, Miguel; Bonete, Rafael; Muñoz de Bustillo, Rafael: *El programa ERASMUS en el marco del valor económico de la Enseñanza del Español como Lengua Extranjera.*
- **DT 06/07** Criado, María Jesús: Inmigración y población latina en los Estados Unidos: un perfil socio-demográ*fico.*
- **DT 05/07** Gutiérrez, Rodolfo: *Lengua, migraciones y mercado de trabajo.*
- **DT 04/07** Quirós Romero, Cipriano; Crespo Galán, Jorge: *Sociedad de la Información y presencia del español en Internet.*
- DT 03/06 Moreno Fernández, Francisco; Otero Roth, Jaime: *Demografía de la lengua española*.
- **DT 02/06** Alonso, José Antonio: *Naturaleza económica de la lengua*.
- **DT 01/06** Jiménez, Juan Carlos: *La Economía de la lengua: una visión de conjunto.*

WORKING PAPERS

- **WP 04/14** Stupariu, Patricia; Vilariño, Angel: *Retos y carencias de la regulación financiera internacional.*
- **WP 03/14** García, Antonio; Molero, José; Rama, Ruth: *Foreign MNEs and domestic innovative capabilities: Are there conditions for reverse spillovers in the spanish industry?*
- **WP 02/14** Sosvilla-Rivero, Simón; Ramos-Herrera, María del Carmen: *On the forecast and consistency of exchange rate expectations: The Spanish PwC Survey.*
- **WP 01/14** Kropacheva, Anna; Molero, José: *Russian tecnological specialization in terms of worls's innovation changes durins 1994-2008. Comparison with countries of BRIC and European Innovation- driven economies.*

WP 07/13	Sanchís, Raúl G.: Extended theory about the allocation of the time. Description and application to the increase in the retirement age policies.
WP 06/13	Morales-Zumaquero, Amalia; Sosvilla-Rivero, Simón: <i>Real exchange rate volatility, financial crises and nominal exchange regimes.</i>
WP 05/13	Álvarez, Isabel; Labra, Romilio: Identifying the role of natural resources in knowledge-based strate- gies of development.
WP 04/13	Alonso Gallo, Nuria; Trillo del Pozo, David: <i>La respuesta de la regulación prudencial a la 29 crisis:</i> Basilea II.
WP 03/13	Revuelta, Julio; Alonso, Fernando: Presencia de las multilatinas en Europa. Tipología y estrategia empresarial.
WP 02/13	Nicolau Ibarra, Ignacio: Evolución de la cooperación española en El Salvador.
WP 01/13	Monedero, Juan Carlos; Jerez, Ariel; Ramos, Alfredo; Fernández, Jose Luis: Participación ciudadana y Democracia. Una revisión de las mejores experiencias Iberoamericanas.
WP 05/12	Sanchís, Raúl G.: Trying to escape the Malaise State in the future. A macroecnomic design to hinder another Great Recession which risks the Welfare State.
WP 04/12	Basave Kunhardt, J., Flujos de IED mexicana hacia Europa y presencia de grandes multinacion- ales mexicanas en España. Evidencia empírica y reflexiones teóricas.
WP 03/12	Luengo Escalonilla, F., Gracia Santos, M., Vicent Valverde, L., <i>Productividad y Posicionamiento Esctructural en la industria de bienes de equipo española.</i>
WP 02/12	Alonso (dir.), José A.; Castillo, Alberto; García, Héctor; Ospina, Shirley; Aguirre, Pablo;
	Millán, Natalia; Santander, Guillermo: Estimación de la ayuda española a la infancia: una pro- puesta metodológica.
WP 01/12	Alonso (dir.), José A.; Aguirre, Pablo; Castillo, Alberto: <i>La cooperación al desarrollo y la</i>
	infancia. Apuntes estratégicos para el caso de España.
WP 09/11	Torrecillas, Celia; Fischer, Bruno B.: Technological Attraction of FDI flows in Knowledge-Inten- sive Services: a Regional Innovation System Perspective for Spain.
WP 08/11	Gómez-Puig, Marta; Sosvilla-Rivero, Simón: Causality and contagion in peripheral emu
	public debt markets: a dynamic approach.
WP 07/11	Sosvilla-Rivero, Simón; Ramos-Herrera, María del Carmen: <i>The US Dollar-Euro exchange rate and US-EMU bond yield differentials: A Causality Analysis.</i>
WP 06/11	Sosvilla-Rivero, Simón; Morales-Zumaquero, Amalia: Volatility in EMU sovereign bond yields: Per- manent and transitory components .
WP 05/11	Castellacci, Fulvio; Natera, José Miguel: A new panel dataset for cross-country analyses of national systems, growth and development (CANA).
WP 04/11	Álvarez, Isabel; Marín, Raquel; Santos-Arteaga, Franciso J.: FDI entry modes, development and technological spillovers.
WP 03/11	Luengo Escalonilla, Fernando: Industria de bienes de equipo: Inserción comercial y cambio estruc- tural.
WP 02/11	Álvarez Peralta, Ignacio; Luengo Escalonilla, Fernando: <i>Competitividad y costes laborales en la UE: más allá de las apariencias.</i>
WP 01/11	Fischer, Bruno B; Molero, José: Towards a Taxonomy of Firms Engaged in International R&D Coop-

eration Programs: The Case of Spain in Eureka.

- **WP 09/10** Éltető, Andrea: Foreign direct investment in Central and East European Countries and Spain a short overview.
- **WP 08/10** Alonso, José Antonio; Garcimartín, Carlos: *El impacto de la ayuda internacional en la calidad de las instituciones.*
- **WP 07/10** Vázquez, Guillermo: Convergencia real en Centroamérica: evidencia empírica para el período 1990-2005.
- **WP 06/10** P. Jože; Kostevc, Damijan, Črt; Rojec, Matija: *Does a foreign subsidiary's network status affect its innovation activity? Evidence from post-socialist economies.*
- **WP 05/10** Garcimartín, Carlos; Rivas Luis; García Martínez, Pilar: On the role of relative prices and capital flows in balance-of-payments constrained growth: the experiences of Portugal and Spain in the euro area.
- **WP 04/10** Álvarez, Ignacio; Luengo, Fernando: *Financiarización, empleo y salario en la UE: el impacto de las nuevas estrategias empresariales.*
- **WP 03/10** Sass, Magdolna: Foreign direct investments and relocations in business services what are the locational factors? The case of Hungary.
- **WP 02/10** Santos-Arteaga, Francisco J.: *Bank Runs Without Sunspots*.
- **WP 01/10** Donoso, Vicente; Martín, Víctor: *La sostenibilidad del déficit exterior de España.*
- **WP 14/09** Dobado, Rafael; García, Héctor: *Neither so low nor so short! Wages and heights in eighteenth and early nineteenth centuries colonial Hispanic America.*
- WP 13/09 Alonso, José Antonio: Colonisation, formal and informal institutions, and development.
- **WP 12/09** Álvarez, Francisco: *Opportunity cost of CO2 emission reductions: developing vs. developed* economies.
- **WP 11/09** J. André, Francisco: *Los Biocombustibles. El Estado de la cuestión.*
- **WP 10/09** Luengo, Fernando: *Las deslocalizaciones internacionales. Una visión desde la economía crítica.*
- **WP 09/09** Dobado, Rafael; Guerrero, David: *The Integration of Western Hemisphere Grain Markets in the Eighteenth Century: Early Progress and Decline of Globalization.*
- **WP 08/09** Álvarez, Isabel; Marín, Raquel; Maldonado, Georgina: *Internal and external factors of competitiveness in the middle-income countries.*
- **WP 07/09** Minondo, Asier: *Especialización productiva y crecimiento en los países de renta media.*
- **WP 06/09** Martín, Víctor; Donoso, Vicente: *Selección de mercados prioritarios para los Países de Renta Media.*
- **WP 05/09** Donoso, Vicente; Martín, Víctor: *Exportaciones y crecimiento económico: estudios empíricos.*
- **WP 04/09** Minondo, Asier; Requena, Francisco: ¿Qué explica las diferencias en el crecimiento de las exportaciones entre los países de renta media?
- **WP 03/09** Alonso, José Antonio; Garcimartín, Carlos: *The Determinants of Institutional Quality. More on the Debate.*
- **WP 02/09** Granda, Inés; Fonfría, Antonio: *Technology and economic inequality effects on international trade.*
- **WP 01/09** Molero, José; Portela, Javier y Álvarez Isabel: *Innovative MNEs' Subsidiaries in different domestic environments.*

WP 08/08	Boege, Volker; Brown, Anne; Clements, Kevin y Nolan Anna: ¿Qué es lo "fallido"? ¿Los Estados del Sur,o la investigación y las políticas de Occidente? Un estudio sobre órdenes políticos híbridos y los Estados emergentes.
WP 07/08	Medialdea García, Bibiana; Álvarez Peralta, Nacho: Liberalización financiera internacional, inver- sores institucionales y gobierno corporativo de la empresa.
WP 06/08	Álvarez, Isabel; Marín, Raquel: FDI and world heterogeneities: The role of absorptive capacities.
WP 05/08	Molero, José; García, Antonio: Factors affecting innovation revisited.
WP 04/08	Tezanos Vázquez, Sergio: The Spanish pattern of aid giving.
WP 03/08	Fernández, Esther; Pérez, Rafaela; Ruiz, Jesús: Double Dividend in an Endogenous Growth Model with Pollution and Abatement.
WP 02/08	Álvarez, Francisco; Camiña, Ester: Moral hazard and tradeable pollution emission permits.
WP 01/08	Cerdá Tena, Emilio; Quiroga Gómez, Sonia: Cost-loss decision models with risk aversion.
WP 05/07	Palazuelos, Enrique; García, Clara: La transición energética en China.
WP 04/07	Palazuelos, Enrique: Dinámica macroeconómica de Estados Unidos: ¿Transición entre dos rece- siones?
WP 03/07	Angulo, Gloria: Opinión pública, participación ciudadana y política de cooperación en España.
WP 02/07	Luengo, Fernando; Álvarez, Ignacio: Integración comercial y dinámica económica: España ante el reto de la ampliación.
WP 01/07	Álvarez, Isabel; Magaña, Gerardo: ICT and Cross-Country Comparisons: A proposal of a new compos- ite index.
WP 05/06	Schünemann, Julia: Cooperación interregional e interregionalismo: una aproximación social-con- structivista.
WP 04/06	Kruijt, Dirk: América Latina. Democracia, pobreza y violencia: Viejos y nuevos actores.
WP 03/06	Donoso, Vicente; Martín, Víctor: <i>Exportaciones y crecimiento en España (1980-2004): Cointe-</i> gración y simulación de Montecarlo.
WP 02/06	García Sánchez, Antonio; Molero, José: Innovación en servicios en la UE: Una aproximación a la den- sidad de innovación y la importancia económica de los innovadores a partir de los datos agregados de la CIS3.
WP 01/06	Briscoe, Ivan: Debt crises, political change and the state in the developing world.
WP 06/05	Palazuelos, Enrique: Fases del crecimiento económico de los países de la Unión Europea–15.
WP 05/05	Leyra, Begoña: Trabajo infantil femenino: Las niñas en las calles de la Ciudad de México.
WP 04/05	Álvarez, Isabel; Fonfría, Antonio; Marín Raquel: <i>The role of networking in the competitive-ness pro-</i> file of Spanish firms.
WP 03/05	Kausch, Kristina; Barreñada, Isaías: <i>Alliance of Civilizations. International Security and Cosmopoli</i> tan Democracy.
WP 02/05	Sastre, Luis: An alternative model for the trade balance of countries with open economies: the Span- ish case.
WP 01/05	Díaz de la Guardia, Carlos; Molero, José; Valadez, Patricia: International competitiveness in services in some European countries: Basic facts and a preliminary attempt of interpreta-tion.
WP 03/04	Angulo, Gloria: La opinión pública española y la ayuda al desarrollo.
WP 02/04	Freres, Christian; Mold, Andrew: European Union trade policy and the poor. Towards im-proving

the poverty impact of the GSP in Latin America.

WP 01/04 Álvarez, Isabel; Molero, José: *Technology and the generation of international knowledge spillovers. An application to Spanish manufacturing firms.*

POLICY PAPERS

Monedero J.C., Democracia y Estado en Améríca Latina: Por una imprudente reinvención de la PP 01/11 política. PP 02/10 Alonso, José Antonio; Garcimartín, Carlos; Ruiz Huerta, Jesús; Díaz Sarralde, Santiago: Strengthening the fiscal capacity of developing countries and supporting the international fight against tax evasión. PP 02/10 Alonso, José Antonio; Garcimartín, Carlos; Ruiz Huerta, Jesús; Díaz Sarralde, Santiago: Fortalecimiento de la capacidad fiscal de los países en desarrollo y apoyo a la lucha internacional contra la evasión fiscal. PP 01/10 Molero, José: Factores críticos de la innovación tecnológica en la economía española. PP 03/09 Ferguson, Lucy: Analysing the Gender Dimensions of Tourism as a Development Strategy. PP 02/09 Carrasco Gallego ,José Antonio: La Ronda de Doha y los países de renta media. PP 01/09 Rodríguez Blanco, Eugenia: Género, Cultura y Desarrollo: Límites y oportunidades para el cambio cultural pro-igualdad de género en Mozambique. PP 04/08 Tezanos, Sergio: Políticas públicas de apoyo a la investigación para el desarrollo. Los casos de Canadá, Holanda y Reino Unido. PP 03/08 Mattioli, Natalia Including Disability into Development Cooperation. Analysis of Initiatives by National and International Donors. PP 02/08 Elizondo, Luis: Espacio para Respirar: El humanitarismo en Afganistán (2001-2008). PP 01/08 Caramés Boada, Albert: Desarme como vínculo entre seguridad y desarrollo. La reintegración comunitaria en los programas de Desarme, desmovilización y reintegración (DDR) de combatientes en Haití. PP 03/07 Guimón, José: Government strategies to attract R&D-intensive FDI. PP 02/07 Czaplińska, Agata: Building public support for development cooperation. PP 01/07 Martínez, Ignacio: La cooperación de las ONGD españolas en Perú: hacia una acción más estratégiса. PP 02/06 Ruiz Sandoval, Erika: Latinoamericanos con destino a Europa: Migración, remesas y codesa-rrollo como temas emergentes en la relación UE-AL. **PP 01/06** Freres, Christian; Sanahuja, José Antonio: Hacia una nueva estrategia en las relaciones Unión Europea – América Latina. PP 04/05 Manalo, Rosario; Reyes, Melanie: The MDGs: Boon or bane for gender equality and wo-men's rights? PP 03/05 Fernández, Rafael: Irlanda y Finlandia: dos modelos de especialización en tecnologías avan-zadas. PP 02/05 Alonso, José Antonio; Garcimartín, Carlos: Apertura comercial y estrategia de desarrollo. PP 01/05 Lorente, Maite: Diálogos entre culturas: una reflexión sobre feminismo, género, desarrollo y mujeres indígenas kichwuas. Álvarez, Isabel: La política europea de I+D: Situación actual y perspectivas. PP 02/04